\documentclass{article}
\usepackage{amsmath,amsfonts}
%\usepackage[OT4,OT1,T1]{fontenc}
%\usepackage[latin1,cp1255]{inputenc}
\usepackage{array}
\makeatletter
%\usepackage{babel}
%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }
%\font\scf=ptmr8r scaled 910
%\font\kukua=ptmr8r scaled 1370
%\font\kukub=ptmr8r scaled 1200
%\usepackage{ij}
%\usepackage{times}
\newcommand{\rF}{\operatorname{F}}
\newcommand{\rG}{\operatorname{G}}
\newcommand{\rSp}{\operatorname{Sp}}
\newcommand{\bbR}{\mathbb{R}}

\textheight=550pt
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=9pt

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}


%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
%{\halign {\normalbaselines%
%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
%&\hbox to 1.3cm{\hfill ##}\cr%
%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
%\def\indexauthor{}
%\def\endindexitem{}
%\def\indexpage{}
%\def\indextitle{}
\newcommand{\indexitem}[3]{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexsee\indexauthor#1\partner#2\endindexitem
%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\makeatother

\begin{document}
\pagestyle{empty}

\index
\indexvolume 160
\indexyear 2007
\endindextitle

%3396 1
\indexitem{Bagno, E. and R.~Biagioli}{Colored-descent representations of complex reflection groups $G(r,p,n)$}{317}

%3416 2
\indexitem{Berele, A.}{Approximate values of the multiplicities in the arm of the cocharacter sequence of $M_{2,1}$}{371}

%3396 3
\indexsee{Biagioli, R.}{E.~Bagno}

%3403 4
\indexitem{Buckner, J. and M.~Dugas}{Quasi-localizations of $\mathbb{Z}$}{349}

%3409 5
\indexitem{Dinur, I., E.~Friedgut, G.~Kindler and R.~O'Donnell}{On the Fourier tails of bounded functions over the discrete cube}{389}

%3403 6
\indexsee{Dugas, M.}{J.~Buckner}

%3409 7
\indexsee{Friedgut, E.}{I.~Dinur}

%3384 8
\indexitem{Helm, D.}{On maps between modular Jacobians and Jacobians of\break Shimura curves}{61}

%3378 9
\indexitem{Hirayama, M. and Y.~Pesin}{Non-absolutely continuous foliations}{173}

%3409 10
\indexsee{Kindler, G.}{I.~Dinur}

%3399 11
\indexitem{Kowalski, E.}{Exponential sums over definable subsets of finite fields}{219}

%3410 12
\indexitem{Ledrappier, F. and O.~Sarig}{Invariant measures for the horocycle flow on periodic hyperbolic surfaces}{281}

%3415 13
\indexitem{Magazanik, E. and M.~A.~Perles}{Relatively convex subsets of simply connected planar sets}{143}

%3414 14
\indexitem{Magazanik, E. and M.~A.~Perles}{Generalized convex kernels of\break simply connected $L_n$ sets in the plane}{157}

%3405 15
\indexitem{Mattarei, S.}{The orders of nonsingular derivations of Lie algebras of characteristic two}{23}

%3387 16
\indexitem{Merimovich, C.}{Prikry on extenders, revisited}{253}

%3409 17
\indexsee{O'Donnell, R.}{I.~Dinur}

%3415 18,19
\indexsee{Perles, M.~A.}{E.~Magazanik}

%3378 20
\indexsee{Pesin, Y.}{M.~Hirayama}

%3386 21
\indexitem{Phillips, N.~C.}{Examples of different minimal diffeomorphisms giving the same C*-algebras}{189}

%3407 22
\indexitem{Rubinstein, J. and I.~Shafrir}{The distance between homotopy\break classes of $S^1$-valued maps in multiply connected
domains}{41}

%3410 23
\indexsee{Sarig, O.}{F.~Ledrappier}

%3407 24
\indexsee{Shafrir, I.}{J.~Rubinstein}

%3393 25
\indexitem{Shilon, O. and B.~Weiss}{Universal minimal topological dynamical systems}{119}

%3393 26
\indexsee{Weiss B.}{O.~Shilon}

%3383 27
\indexitem{Xi, L.-F.}{Quasi-Lipschitz equivalence of fractals}{1}

\end{document}